Author's response to reviews

Title: Use of an automated blood culture system (BD BACTECTM) for diagnosis of prosthetic joint infections: easy and fast

Authors:

   Angela M Minassian (angela.minassian@ouh.nhs.uk)
   Robert Newnham (robert.newnham@ouh.nhs.uk)
   Elizabeth Kalimeris (elizabeth.kalimeris@ouh.nhs.uk)
   Philip Bejon (pbejon@well.ox.ac.uk)
   Bridget L Atkins (bridget.atkins@ouh.nhs.uk)
   Ian CJW Bowler (ian.bowler@ouh.nhs.uk)

Version: 2
Date: 14 January 2014

Author's response to reviews:

Additional statement included at the end of the methods section:

"This report is an analysis of data from routine laboratory processing of samples taken for routine healthcare. No research related contact with human subjects occurred. Our institution, a National Health Service Hospital Trust, classifies our work as a service evaluation and therefore did not require a submission for ethical review".